Diabetes with Hypertension as Risk Factors for Adult Dengue Hemorrhagic Fever in a Predominantly Dengue Serotype 2 Epidemic: A Case Control Study

Junxiong Pang^{1,2*}, Agus Salim², Vernon J. Lee^{2,3}, Martin L. Hibberd^{1,2}, Kee Seng Chia², Yee Sin Leo^{4,5}, David C. Lye^{4,5}

1 Infectious Diseases, Genome Institute of Singapore, Singapore, 2 Saw Swee Hock School of Public Health, National University of Singapore, Singapore, Singapore, 3 Department of Clinical Epidemiology, Tan Tock Seng Hospital, Singapore, Singapore, 4 Department of Infectious Diseases, Tan Tock Seng Hospital, Singapore, Singapore, 5 Department of Medicine, Yong Loo Lin School of Medicine, National University of Singapore, Singapore

Abstract

Background: Dengue hemorrhagic fever (DHF) is a severe form of dengue, characterized by bleeding and plasma leakage. A number of DHF risk factors had been suggested. However, these risk factors may not be generalized to all populations and epidemics for screening and clinical management of patients at risk of developing DHF. This study explored demographic and comorbidity risk factors for DHF in adult dengue epidemics in Singapore in year 2006 (predominantly serotype 1) and in year 2007–2008 (predominantly serotype 2).

Methods: A retrospective case-control study was conducted with 149 DHF and 326 dengue fever (DF) patients from year 2006, and 669 DHF and 1,141 DF patients from year 2007–2008. Demographic and reported comorbidity data were collected from patients previously. We performed multivariate logistic regression to assess the association between DHF and demographic and co-morbidities for year 2006 and year 2007–2008, respectively.

Results: Only Chinese (adjusted odds ratio [AOR] = 1.90; 95% confidence interval [Cl]: 1.01–3.56) was independently associated with DHF in year 2006. In contrast, age groups of 30–39 years (AOR = 1.41; 95% Cl:1.09–1.81), 40–49 years (AOR = 1.34; 95% Cl:1.09–1.81), female (AOR = 1.57; 95% Cl:1.28–1.94), Chinese (AOR = 1.67; 95% Cl:1.24–2.24), diabetes (AOR = 1.78; 95% Cl:1.06–2.97), and diabetes with hypertension (AOR = 2.16; 95%Cl:1.18–3.96) were independently associated with DHF in year 2007–2008. Hypertension was proposed to have effect modification on the risk of DHF outcome in dengue patients with diabetes. Chinese who had diabetes with hypertension had 2.1 (95% Cl:1.07–4.12) times higher risk of DHF compared with Chinese who had no diabetes and no hypertension.

Conclusions: Adult dengue patients in Singapore who were 30–49 years, Chinese, female, had diabetes or diabetes with hypertension were at greater risk of developing DHF during epidemic of predominantly serotype 2. These risk factors can be used to guide triaging of patients who require closer clinical monitoring and early hospitalization in Singapore, when confirmed in more studies.

Citation: Pang J, Salim A, Lee VJ, Hibberd ML, Chia KS, et al. (2012) Diabetes with Hypertension as Risk Factors for Adult Dengue Hemorrhagic Fever in a Predominantly Dengue Serotype 2 Epidemic: A Case Control Study. PLoS Negl Trop Dis 6(5): e1641. doi:10.1371/journal.pntd.0001641

Editor: Scott B. Halstead, Pediatric Dengue Vaccine Initiative, United States of America

Received November 23, 2011; Accepted March 28, 2012; Published May 1, 2012

Copyright: © 2012 Pang et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by the National Medical Research Council grant NMRC/TCR/005/2008. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have read the journal's policy and have the following conflicts: VJL has received unrelated research funding from GlaxoSmithKline; YSL is a consultant to Sanofi-Pasteur.

* E-mail: pangjx99@gis.a-star.edu.sg

Introduction

Dengue is a major neglected tropical disease in the tropical and subtropical regions of the world [1]. It is predominantly found in urban and semi-urban areas, and results in a wide spectrum of clinical manifestations, from asymptomatic infection, undifferentiated fever, dengue fever (DF) to severe infection known as dengue hemorrhagic fever (DHF) and dengue shock syndrome (DSS) [2]. It is estimated about 50 million infections occur annually, with 500,000 DHF cases and 22,000 deaths [3]. In several Asian countries, dengue is one of the leading causes of hospitalization and death among children [1–3]. In Singapore, however, there is a decreasing trend of children (aged <15 years) and an increasing trend of adults (aged ≥ 25 years) being infected with dengue since 1982 [4].

Dengue hemorrhagic fever is characterized by bleeding and plasma leakage which may lead to life-threatening shock, if unrecognized and not treated in a timely manner. Molecular determinants of DHF such as virus variation, viral load, antibodydependent enhancement (ADE), 'original antigenic sin', 'cytokine storm' and plasma factors were proposed in the pathophysiology of DHF [5–7]. However, predicting or preventing the occurrence of DHF remains a challenge. Identifying risk factors for DHF can facilitate early clinical, preventive and healthcare resource

Author Summary

Dengue is a major vector borne disease in the tropical and subtropical regions. An estimated 50 million infections occur per annum in over 100 countries. A severe form of dengue, characterized by bleeding and plasma leakage, known as dengue hemorrhagic fever (DHF) is estimated to occur in 1-5% of hospitalized cases. It can be fatal if unrecognized and not treated in a timely manner. Previous studies had found a number of risk factors for DHF. However, screening and clinical management strategies based on these risk factors may not be applicable to all populations and epidemics of different serotypes. In this study, we found significant association between DHF and diabetes mellitus and diabetes mellitus with hypertension during the epidemic of predominantly serotype 2 (year 2007 and 2008), but not during the epidemic of predominantly serotype 1 (year 2006). Diabetes mellitus and hypertension are prevalent in Singapore and most parts of South-East Asia, where dengue is endemic. Therefore, it is important to address the risk effect of these co-morbidities on the development of DHF so as to reduce morbidity and mortality. Our findings may have impact on screening and clinical management of dengue patients, when confirmed in more studies.

management. Epidemiological risk factors of DHF such as dengueserotype 2 [8,9], Asian genotype [10], prior dengue infections [11,12], children [13,14], age >65 years [15], white females [12,16] were identified. Integrative analysis of these risk factors, together with the molecular determinants of DHF, may facilitate better understanding of the pathophysiology of DHF.

Co-morbidities were reported as risk factors for DHF in a number of studies from dengue endemic countries. These co-morbidities included sickle cell anemia [17], asthma [17–19], hypertension [15,16,18], uremia [15], allergies treated with corticosteroid [16] and diabetes mellitus [15–18]. However, these co-morbidities may not be generalized to all populations and epidemics of all dengue serotypes. Furthermore, most of these risk factors were identified from univariate analysis [15,17–19] instead of multivariate analysis to adjust for potential confounders [16]. In this study, we explored demographic and co-morbidity risk factors for DHF in Singapore in the year 2006 (where dengue serotype 1 predominated) and in the year 2007 and 2008 (where dengue serotype 2 predominated) [20].

Methods

A retrospective case-control study was conducted using data collected from all adult dengue patients admitted from 1 January 2006 to 31 December 2008 to the Department of Infectious Diseases at Tan Tock Seng Hospital (TTSH). It is the largest hospital in Singapore for the treatment of dengue patients where dengue patients were managed using a standardized dengue care path. The extracted data was de-identified in analysis. Dengue fever was defined as reported fever or measured temperature of >38°C with any two of the following manifestations (according to WHO 1997 criteria [2]): headache, retro-orbital pain, myalgia, arthralgia, rash, hemorrhagic manifestations, or leukopenia. Probable dengue patients had positive acute dengue serology, as measured by Dengue Duo IgM & IgG Rapid Strip Test (Panbio Diagnostic, Queensland, Australia) [21,22], and clinical diagnostic criteria of dengue fever by WHO 1997 [2]. The IgG test line for this Rapid Strip Test is set to detect the high levels of IgG characteristic of secondary virus infection, and has been validated to detect acute secondary infection [22]. Confirmed dengue patients had positive dengue polymerase chain reaction (PCR) assay [23] and clinical diagnostic criteria according to WHO 1997 [2]. Dengue hemorrhagic fever was diagnosed when all four criteria of fever, hemorrhagic manifestations, thrombocytopenia ($<100 \times 10^9/L$) and evidence of plasma leakage (hematocrit change $\geq 20\%$, hypoproteinemia or clinical fluid accumulation) were present [2].

Patients with outcome diagnosed as DHF and DF were classified as the case group and control group respectively. In order to compare the demographic and co-morbidity profiles between DHF and DF patients, data were extracted from chart review. Demographic data extracted were age, gender and ethnicity. Comorbidities data extracted were diabetes mellitus, hypertension, asthma, hyperlipidemia, stroke, chronic obstructive pulmonary disease, corticosteroid use and HIV/AIDS. However, stroke, chronic obstructive pulmonary disease, corticosteroid use and HIV/AIDS were not included in the analysis because these comorbidities were either not reported or reported by very few cases (DHF) or controls (DF). Patients with diabetes mellitus tend to have other co-morbidities, and the risk effect of diabetes mellitus with an additional co-morbidity on DHF outcome was also investigated.

Statistical methods

For descriptive analysis, Pearson's chi-square and Fisher's exact tests were used to compare categorical variables, and Mann-Whitney U test was used to compare continuous variables with nonnormal distribution. Univariate and multivariate logistic regression were used to calculate crude and adjusted odds ratios (COR, AOR), respectively, and their 95% confidence intervals (CI) were used to assess the association of the variables with DHF. Confounding effect was minimized by performing multivariate logistic regression adjusting for potential confounders identified in the descriptive analysis. These potential confounders are exposures that were found to be statistically different (p<0.05) between DHF and DF patients in Table 1 and the model that fits best in the multivariate regression is tested using likelihood-ratio test. In Singapore, dengue infections were predominantly due to dengue serotype 1 (detected in 75% to 100% of dengue samples collected each month) during the epidemic in the year 2006, and dengue serotype 2 (detected in up to 91% of dengue samples) during the epidemic in the year 2007 and 2008 [20]. Given that different dengue serotypes may cause different disease severity [24], the data from year 2006 and the data from year 2007 to 2008 were analysed separately to minimize confounding effect due to different predominant dengue serotypes. Stratified analyses were performed to evaluate the presence of effect modification between diabetes mellitus and other co-morbidities on the risk of DHF outcome. All statistical analyses were performed using Stata 10.0 (Stata Corp., College Station, TX, 2005). All tests were conducted at the 5% level of significance, with OR, P-value and corresponding 95% CI reported where applicable.

Ethics approval

This study was approved by Domain Specific Review Board, National Healthcare Group, Singapore (DSRB-E/08/567) with waiver of informed consent as this was a retrospective study and the data were analyzed anonymously.

Results

Demographic and co-morbidity profiles of DHF & DF patients during the two epidemics

In year 2006 epidemic, there were 149 DHF and 326 DF patients. Among these patients, there were 131 (27.6%) patients

Table 1. Distribution of demographic characteristics and co-morbidities by cases (DHF) and controls (DF) in year 2006 and year 2007–2008 epidemics.

Year Exposure	2006 (N = 475)						2007–2008 (N = 1810)					
	Cases (N = 149)		Controls (N = 326)		Case : Control	Cases (N = 669)		Controls (N = 1141)		Case : Control		
	N	%	Ν	%		\mathbf{P} -value $^{\Delta}$	Ν	%	Ν	%		\mathbf{P} -value ^{Δ}
Age (Years)												
Mean (SD)	37.3	(12.8)	34.0	(11.0)		0.008 ^	38.4	(13.4)	36.2	(12.9)		<0.001 ^
<30	43	28.9	115	35.3	1 : 2.67		180	26.9	405	35.5	1 : 2.25	
30–39	46	30.9	127	39.0	1 : 2.76		209	31.2	328	28.8	1:1.57	
40–49	39	26.2	58	17.8	1 : 1.49		152	22.7	213	18.7	1:1.40	
50–59	12	8.1	19	5.8	1 : 1.58		76	11.4	132	11.6	1:1.74	
≥60	9	6.0	7	2.2	1:0.78	0.017	52	7.8	63	5.5	1:1.21	0.002
Gender												
Male	101	67.8	233	71.5	1:2.31		393	58.7	809	70.9	1 : 2.06	
Female	48	32.2	93	28.5	1:1.94	0.414	276	41.3	332	29.1	1 : 1.20	<0.001
Ethnicity												
Others	15	10.1	60	18.4	1:4.00		79	11.8	188	16.5	1 : 2.38	
Chinese	115	77.2	205	62.9	1:1.78		516	77.1	717	62.8	1:1.39	
Malay	4	2.7	12	3.7	1:3.00		38	5.7	62	5.4	1 : 1.63	
Indian	15	10.1	49	15.0	1:3.27	0.021	36	5.4	174	15.3	1:4.83	<0.001
Fever DBP												
Mean (SD)	4.9	(1.5)	4.9	(1.6)		0.941^	4.9	(1.7)	5.0	(1.8)		0.308^
Hypertension												
No	140	94.0	315	96.6	1 : 2.25		594	88.8	1047	91.8	1 : 1.76	
Yes	9	6.0	11	3.4	1 : 1.22	0.179	75	11.2	94	8.2	1:1.25	0.036
Diabetes												
No	147	98.7	319	97.9	1:2.17		626	93.6	1101	96.5	1 : 1.76	
Yes	2	1.3	7	2.2	1:3.50	0.726 [#]	43	6.4	40	3.5	1:0.93	0.004
Hyperlipidemia												
No	145	97.3	317	97.2	1:2.19		612	91.5	1061	93.0	1:1.73	
Yes	4	2.7	9	2.8	1 : 2.25	1.000#	57	8.5	80	7.0	1 : 1.40	0.241
Asthma												
No	145	97.3	311	95.4	1 : 2.15		637	95.2	1082	94.8	1 : 1.70	
Yes	4	2.7	15	4.6	1:3.75	0.323	32	4.8	59	5.2	1:1.84	0.740

 ${}^{\Delta}\textsc{Person's}$ Chi-square, unless otherwise annotated.

Mann-Whitney U test.

#Fisher's Exact test.

DHF-Dengue hemorrhagic fever.

DF- Dengue fever.

DBP- Days before presentation in hospital.

doi:10.1371/journal.pntd.0001641.t001

who were PCR positive and 344 (72.4%) patients who were serology positive but PCR negative. The mean age was 37.3 (\pm 12.8) years and 34.0 (\pm 11.0) years for DHF patients and DF patients respectively. Among DHF patients, there were 67.8% male and 77.2% Chinese. Of the 326 DF patients, 71.5% were male and 62.9% were Chinese (Table 1). In year 2007 and 2008 epidemic, there were 669 DHF and 1,141 DF patients. Among these patients, there were 590 (32.6%) patients who were PCR positive and 1220 (67.4%) patients who were serology positive but PCR negative. The mean age was 38.4 (\pm 13.4) years and 36.2 (\pm 12.9) years for DHF patients and DF patients respectively. Among the DHF patients, there were 58.7% male and 77.1% Chinese. Of the 1,141 DF patients, 70.9% were male and 62.8% were Chinese (Table 1). Of the demographic variables, statistically significant differences (P<0.05) were found between DHF and DF with respect to mean age (P=0.008), age groups (P=0.017) and ethnicity (P=0.021) in year 2006, and mean age (P<0.001), age groups (P=0.002), gender (P<0.001) and ethnicity (P<0.001) in year 2007 and 2008 (Table 1). Using the number of fever days before hospital presentation as a surrogate index of health-seeking behavior between DHF and DF patients, no significant difference was observed in both year 2006 (P=0.941) as well as year 2007–2008 (P=0.308) (Table 1). Notably, statistically significant differences were found between DHF and DF with respect to hypertension (P=0.036) and diabetes mellitus (P=0.004) in year 2007 and 2008 but not year 2006 (Table 1).

Independent risk factors for DHF

Chinese ethnicity was the only significant risk factor independently associated with DHF in year 2006, after adjustment for statistically significant univariate risk factors (Table 2). Although marginally significant, the likelihood (AOR) of a Chinese patient developing DHF was 1.90 (95% CI:1.01-3.56) times higher than that of other ethnicity (not Chinese, Malay or Indian). In year 2007 and 2008, age groups, gender and ethnicity were observed to be independently associated with DHF, following adjustment for statistically significant univariate risk factors (Table 2). The likelihood (AOR) of an individual who were 30 to 39 years of age and 40 to 49 years of age developing DHF was 1.41 (95% CI:1.09-1.81) and 1.34 (95% CI:1.09-1.81) times higher than that of an individual below 30 years old respectively. Females had 1.57 (95% CI:1.28–1.94) times higher risk developing DHF than males. In addition, the likelihood (AOR) of a Chinese patient developing DHF was 3.15 (95% CI:2.34-4.23) and 1.67 (95% CI:1.24-2.24) times higher than that of Indian and other ethnicity respectively (Table 2).

For co-morbidities, after adjustment for statistically significant univariate risk factors, only diabetes mellitus remained an independent risk factor for DHF outcome (AOR = 1.78; 95% CI:1.06–2.97) in year 2007 and 2008 (Table 3). Diabetic patients tend to have other co-morbidities. We investigated the risk effect of DHF outcome on patients having diabetes mellitus with hypertension, hyperlipidemia or asthma. Diabetes mellitus with hypertension (COR = 2.43; 95% CI:1.42–4.15), diabetes mellitus with hyperlipidemia (COR = 1.82; 95% CI:1.06–3.12) and diabetes mellitus with no asthma (COR = 1.74; 95% CI:1.10–2.76) were observed to be significantly associated with DHF outcome (Table 4). However, only diabetes mellitus with hypertension (AOR = 2.16; 95% CI:1.18–3.96) and diabetes with no asthma (AOR = 1.68; 95% CI:1.02–2.76) were observed to be independently associated with DHF outcome after adjustment for statistically significant univariate risk factors (Table 4). Interestingly, the likelihood (AOR) of an individual having diabetes mellitus with asthma developing DHF was 4.38 (95% CI:0.80-23.85) times higher than that of an individual having no diabetes with no asthma. However, there is a lack of statistical significance and it is most likely due to the small sample size with only 7 subjects having diabetes mellitus with asthma (Table 4). In order to confirm this observed phenomenon, further studies with larger sample size are required. In addition, among patients with hypertension, the likelihood (AOR) of developing DHF due to diabetes mellitus was higher (AOR = 2.39; 95% CI:1.21-4.71) compared to that of patients without hypertension (AOR = 1.28; 95% CI:0.56-2.93; Table 5). This provided preliminary evidence of effect modification between diabetes mellitus and hypertension on the risk of DHF outcome. Moreover, it was observed that the mean hospitalization days was longer for diabetic patients $(4.99\pm3.34 \text{ days})$ compared to non-diabetic patients (4.04 ± 1.62) days; P = 0.001). Significant difference was also observed in the mean hospitalization days between diabetic DHF patients and non-diabetic DHF patients (diabetic DHF: 5.21±3.12 days; nondiabetic DHF: 4.33 ± 1.75 days; P = 0.046) (data not shown).

Subgroup analyses of patients with dengue IgG data and of Chinese patients were carried out. In the subgroup analysis of 1,220 (67.4%) patients hospitalized during the year 2007–2008 that had dengue IgG data, we further showed that diabetes (AOR: 1.92; 95% CI: 1.02–3.61) as well as diabetes with hypertension (AOR: 4.41; 95% CI: 1.16–16.82) remained as risk factor of DHF (Table S1). Furthermore, in a subgroup analysis of cases (DHF) and controls (DF) identified as Chinese in year 2007 and 2008, diabetes mellitus (AOR = 2.23; 95% CI:1.21–4.11), diabetes mellitus with hypertension (AOR = 2.1; 95% CI:1.07–4.12), diabetes mellitus with no hyperlipidemia (AOR = 3.75; 95%

Table 2. Crude and adjusted odds ratios of the association of DHF with age groups, gender and ethnicity in year 2006 and year 2007–2008 epidemics.

Year	2006				2007-2008	8		
Exposure	COR	95% CI	AOR*	95% CI	COR	95% CI	AOR**	95% CI
Age (Years)								
<30	1		1		1		1	
30–39	0.97	0.60-1.58	0.92	0.56-1.50	1.43	1.12-1.84	1.41	1.09-1.81
40–49	1.80	1.05-3.07	1.53	0.88-2.67	1.61	1.22-2.11	1.34	1.09–1.81
50–59	1.69	0.76-3.77	1.47	0.65-3.31	1.30	0.93-1.81	0.91	0.63-1.30
≥60	3.44	1.20-9.81	2.71	0.94-7.88	1.86	1.24-2.79	1.14	0.70-1.85
Gender								
Male	1		1		1		1	
Female	1.19	0.78-1.81	1.14	0.74-1.78	1.71	1.40-2.09	1.57	1.28-1.94
Ethnicity								
Others	1		1		1		1	
Chinese	2.24	1.22-4.13	1.90	1.01-3.56	1.71	1.29-2.28	1.67	1.24-2.24
Malay	1.33	0.38-4.72	1.15	0.32-4.13	1.46	0.90-2.36	1.30	0.79–2.13
Indian	1.22	0.55-2.75	1.19	0.53-2.68	0.49	0.32-0.77	0.53	0.34-0.83

*Adjusted odds ratio was obtained from a multivariate logistic regression being adjusted by age groups and ethnicity.

**Adjusted odds ratio was obtained from a multivariate logistic regression being adjusted by age groups, gender, ethnicity, diabetes mellitus and hypertension. DHF- Dengue Hemorrhagic Fever.

COR- Crude odds ratio.

AOR- Adjusted odds ratio.

CI- Confidence interval.

doi:10.1371/journal.pntd.0001641.t002

 Table 3. Crude and adjusted odds ratios of the association of DHF with co-morbidities in year 2006 and year 2007–2008

 epidemics.

Year	2006				2007-200	8		
Exposure	COR	95% CI	AOR*	95% CI	COR	95% CI	AOR**	95% CI
Hypertension								
No	1		1		1		1	
Yes	1.84	0.74-4.54	0.97	0.31-3.00	1.41	1.02–1.94	1.06	0.70-1.60
Diabetes								
No	1		1		1		1	
Yes	0.62	0.13-3.02	0.34	0.06–1.89	1.89	1.21-2.94	1.78	1.06-2.97
Hyperlipidemia								
No	1		1		1		1	
Yes	0.97	0.29-3.20	0.54	0.15-1.96	1.24	0.87-1.76	0.79	0.50-1.26
Asthma								
No	1		1		1		1	
Yes	0.57	0.19-1.75	0.51	0.16-1.62	0.92	0.59-1.43	0.86	0.55-1.35

*Adjusted Odds Ratio was obtained from a multivariate logistic regression being adjusted by age groups and ethnicity.

**Adjusted odds ratio was obtained from a multivariate logistic regression being adjusted by age groups, gender, ethnicity, diabetes mellitus and hypertension. DHF- Dengue Hemorrhagic Fever.

COR- Crude odds ratio.

AOR- Adjusted odds ratio.

Cl- Confidence interval.

doi:10.1371/journal.pntd.0001641.t003

Exposures	Cases	Controls				
	N	N	COR	95% CI	AOR*	95% CI
Diabetes						
No	626	1101	1		1	
Yes	43	40	1.89	1.21-2.94	1.78	1.06-2.97
Diabetes, Hypertension						
No diabetes with no hypertension	584	1031	1		1	
No diabetes with hypertension	42	70	1.06	0.71-1.57	0.97	0.62–1.52
Diabetes with no hypertension	10	16	1.1	0.50-2.45	1.26	0.55-2.87
Diabetes with hypertension	33	24	2.43	1.42-4.15	2.16	1.18-3.96
Diabetes, Hyperlipidemia						
No diabetes with no hyperlipidemia	597	1048	1		1	
No diabetes with hyperlipidemia	29	53	0.96	0.60-1.53	0.82	0.50-1.37
Diabetes with no hyperlipidemia	15	13	2.03	0.96-4.29	2.03	0.93-4.47
Diabetes with hyperlipidemia	28	27	1.82	1.06-3.12	1.62	0.90-2.92
Diabetes, Asthma						
No diabetes with no asthma	599	1044	1		1	
No diabetes with asthma	27	57	0.83	0.52-1.32	0.79	0.49-1.27
Diabetes with no asthma	38	38	1.74	1.10-2.76	1.68	1.02-2.76
Diabetes with asthma	5	2	4.36	0.84-22.53	4.38	0.80-23.85

 Table 4. Crude and adjusted odds ratios of the association of DHF with multiple co-morbidities in year 2007–2008 epidemic.

*Adjusted odds ratio was obtained from a multivariate logistic regression being adjusted by age groups, gender, ethnicity, diabetes mellitus and hypertension. DHF- Dengue Hemorrhagic Fever.

COR- Crude odds ratio.

AOR- Adjusted odds ratio.

CI- Confidence interval.

doi:10.1371/journal.pntd.0001641.t004

Table 5. Interaction effect between diabetes mellitus and other comorbidities in year 2007–2008 epidemic.

Exposures	Cases	Controls				
	N	N	COR	95% CI	AOR*	95% CI
Diabetes						
No	626	1101	1		1	
Yes	43	40	1.89	1.21-2.94	1.78	1.06-2.97
No hypertension						
No diabetes	584	1031	1		1	
Diabetes	10	16	1.10	0.50-2.45	1.28	0.56-2.93
Hypertension						
No diabetes	42	70	1		1	
Diabetes	33	24	2.29	1.20-4.39	2.39	1.21-4.71
No hyperlipidemia						
No diabetes	597	1048	1		1	
Diabetes	15	13	2.03	0.96-4.29	1.95	0.88-4.36
Hyperlipidemia						
No diabetes	29	53	1		1	
Diabetes	28	27	1.90	0.94-3.80	2.03	0.93-4.41
No asthma						
No diabetes	599	1044	1		1	
Diabetes	38	38	1.74	1.10-2.76	1.77	1.03-3.00
Asthma						
No diabetes	27	57	1		1	
Diabetes	5	2	5.28	0.96-29.0	1.01	0.04-25.15

*Adjusted odds ratio was obtained from a multivariate logistic regression being adjusted by age groups, gender, ethnicity, diabetes mellitus and hypertension. DHF- Dengue Hemorrhagic Fever.

COR- Crude odds ratio.

AOR- Adjusted odds ratio.

CI- Confidence interval.

doi:10.1371/journal.pntd.0001641.t005

CI:1.27–11.02) and diabetes mellitus with no asthma (AOR = 1.96; 95% CI: 1.09-3.52) were independently associated with DHF outcome, after adjustment for age groups, gender, and hypertension (data not shown).

Discussion

The results of this study showed that female, Chinese, age group between 30 to 49 years, pre-existing diabetes mellitus or diabetes mellitus with hypertension were risk factors of developing DHF during the year 2007 and 2008 epidemic when dengue serotype 2 was predominant. In contrast, Chinese ethnicity was the only risk factor observed during the year 2006 epidemic when dengue serotype 1 was predominant. This might be due to the different predominant circulating dengue serotypes during the two epidemics [20]. Notably, dengue serotype 2 was known to be associated with more severe dengue disease than serotype 1 [8,9,24]. In a combined analysis of year 2006, 2007 and 2008 epidemic, all the risk factors identified in the year 2007-2008 epidemic remained as independent risk factors except for diabetes mellitus (Table S2). This may suggest potential confounding effect of different serotypes. Furthermore, it was observed that age, gender and co-morbidities were not independently associated with DHF outcome in a previous study of 1,973 adult dengue patients in the year 2004 epidemic when dengue serotype 1 was also predominant [25]. However, it is not possible to conclusively

demonstrate serotype difference during epidemics as the main factor that accounted for the differences in risk factors in this study. This was because individual serotype data was inaccessible for our analyses due to national regulations. Instead, the differences in risk factors may be due to the small sample size of patients admitted in year 2006, and the significant differences in mean age, number of patients with co-morbidities and DHF outcome admitted during the two epidemics (Table S3). It is beyond the scope of this study to highlight other potential factors, such as climate change, viral genotype change as well as change in health-seeking behaviors that may have also resulted in the differences.

It is not surprising that female and Chinese ethnicity were independent risk factors of DHF as gender [12,26] and ethnicity [12,16] were shown to be risk factors for DHF in Cuba and Brazil studies as well as in Vietnam for dengue shock syndrome (DSS). Age groups between 30 and 39 and between 40 and 49 were independent risk factors of DHF in our adult dengue cohort. This observation is different from previous studies in Cuba [13] and in Singapore [14] where children <14 years of age had higher risk of developing DHF compared to young adults aged 15 years or greater. The rationale behind this difference could be due to lowered herd immunity and change in transmission pattern [14,27]. The elderly (>65 years of age) in Taiwan [15] had a higher risk of developing DHF. However, the age group \geq 60 year was not an independent risk factor of DHF outcome (Table 2) in our current study which is also consistent with our previous study on dengue in older adults [28]. It is still not well understood how these risk factors contribute to the pathophysiology of DHF, and understanding the underlying mechanism may facilitate clinical management.

Co-morbidities such as hypertension, diabetes mellitus, hyperlipidemia and asthma are among the few leading causes of mortality and morbidity in Asia [29] and globally [30,31]. Comorbidities were shown to be associated with severe clinical manifestations of several infectious disease such as SARS [32,33], pandemic influenza H1N1 [34], tuberculosis [35,36], hepatitis C [37] and community-acquired infections [38,39]. Many studies found association between various co-morbidities and DHF outcome [15-19] but only one study was carried out with multivariate analysis to adjust for potential confounders [16]. Furthermore, none has evaluated the risk effect of two existing comorbidities and the effect modification between two co-morbidities on DHF outcome. In this study, we showed that diabetes mellitus was associated with DHF outcome as observed by other studies [15–18]. In addition, we observed that individuals reported having diabetes mellitus with hypertension had higher risk of developing DHF compared with individuals with no diabetes mellitus and no hypertension. Our study may be the first that provide the preliminary evidence of synergy of risk effect between diabetes mellitus and hypertension on DHF outcome (Tables 4 & 5). Our study showed concomitant diabetes mellitus with hypertension as an independent risk factor for DHF in a large number of adult DHF cases in Singapore, and supported the initial evidence of association between hospitalization with a diagnosis of DHF and diabetes mellitus in Brazil [16]. However, the pathophysiology behind diabetes leading to DHF outcome is not well understood yet, even though numerous studies had suggested that diabetes mellitus can result in immune and endothelial dysfunction [40-44].

Identifying risk factors for DHF can guide clinicians to triage dengue patients for the right site of care for closer monitoring and early intervention with fluid resuscitation. In an epidemic where healthcare resources may be stretched, risk factors for DHF can be used to prioritize hospitalization of dengue patients. In our study, we observed that diabetic patients with DHF outcome required longer stay and, presumably, required more medical attention in the hospital compared to non-diabetic patients with DHF. Additionally, policy makers can prioritize population groups at high risk of developing DHF such as female patients, patients in age group 30-49, and patients having diabetes or diabetes with hypertension for vaccination when dengue vaccines are available, particularly in resource-limited countries. Demographic and comorbidity risk factors may help public health clinicians raise awareness among high-risk individuals to take preventive measures against dengue infections.

As this is a retrospective study, the quality of the study was dependent on the quality of the data available and collected. Information bias was minimized by the use of the standardized dengue care path for consistent clinical documentation. Reporting bias was minimized by the fact that patients with comorbidities tend to know their existing condition and are likely to be on constant medication. However, it is challenging to exclude the fact that there are no undetected existing comorbidities among some of these patients as this study is performed retrospectively. In addition, there may be selection bias because the subjects were all hospitalized and hence were likely to have active health careseeking behaviour, and the controls were hospitalized DF patients who may not truly represent the general population. In the general population, less active health care-seeking, asymptomatic or mild DF patients may not visit a doctor or hospital and may also have diabetes mellitus or other co-morbidities assessed in this study. However, it is technically challenging to identify these less active health care-seeking, asymptomatic or mild DF patients for inclusion in the study. We also did not have patient-specific dengue serotype data and could only extrapolate our observations from previous population study in Singapore [20]. Lastly, we understand the importance of accounting for prior infection as it is a main risk factor for DHF. The result of IgG test carried out within seven days of fever onset can be used to classify patients with or without prior infection [22,45]. However, we only have had IgG results of 67.4% of all patients during the year 2007-2008. In the subgroup analysis, we showed that prior infection was not significantly associated with DHF in adult patients (Table S1). Furthermore, it has been shown that prior infection was strongly associated with DHF in children under the age of 15 years [24,46]. In other words, this may suggest that diabetes as well as diabetes with hypertension may be risk factors of DHF in adults, regardless of prior dengue infections. Further studies involving larger number of patients with acute secondary infections are required to confirm this hypothesis.

In conclusion, we found age between 30 and 49 years, female gender, Chinese ethnicity, diabetes mellitus and diabetes mellitus with hypertension to be independent risk factors for DHF in an adult dengue epidemic with predominantly dengue serotype 2. The two co-morbidities appeared to have effect modification on the risk of DHF outcome. More studies, particularly prospective studies are required to confirm these findings. Our finding raised the likely association between the pathophysiology of diabetes mellitus, hypertension and dengue severity. An ongoing genomewide association study in Singapore may help elucidate genetic predisposition to severe dengue disease including the role of diabetes mellitus.

Supporting Information

Table S1 Subgroup analyses of dengue patients with IgG Rapid Test data in year 2007–2008 epidemic. Subgroup analyses of adult patients with IgG Rapid Test data suggested that female, Chinese, diabetes as well as diabetes with hypertension remain as significant independent risk factors of DHF.

(DOC)

Table S2 Combined analyses of adult dengue patients admitted in year 2006, 2007 and 2008 epidemic. Combined analyses of dengue patients admitted in the three consecutive years showed that age group from 30–49, female, Chinese, but not diabetes, are significant independent risk factors for DHF.

 $\left(\text{DOC} \right)$

Table S3 Descriptive analysis of demographics and comorbidities variables between epidemics in year 2006 and year 2007–2008. There are significant differences in the distribution of age groups, number of adult dengue patients with diabetes, hypertension and hyperlipidemia, as well as the number of DHF cases between the group of patients admitted in year 2006 epidemic and the group admitted in year 2007–2008 epidemic.

Acknowledgments

We are grateful to Ms. Adriana S. Tan and Dr. Wah Wah Lin for data analysis and entry, respectively.

⁽DOC)

Author Contributions

Conceived and designed the experiments: JP AS VJL MLH KSC YSL DCL. Performed the experiments: JP AS VJL DCL . Analyzed the data: JP

References

- World Health Organization (2010) First WHO report on neglected tropical diseases: working to overcome the global impact of neglected tropical diseases. Available: http://whqlibdoc.who.int/publications/2010/9789241564090_eng. pdf. Accessed 2011 Nov 22.
- World Health Organization (1997) Dengue haemorrhagic fever: diagnosis, treatment, prevention and control. Second edition. Available: http://www.who. int/csr/resources/publications/dengue/001-11.pdf. Accessed 2011 Nov 22.
- World Health Organization (2009) Dengue: Guidelines for diagnosis, treatment, prevention and control. Available: http://whqlibdoc.who.int/publications/ 2009/9789241547871_eng.pdf. Accessed 2011 Nov 22.
- Ooi EE, Goh KT, Gubler DJ (2006) Dengue prevention and 35 years of vector control in Singapore. Emerg Infect Dis 12: 887–893.
- Rothman AL (2003) Immunology and immunopathogenesis of dengue disease. Adv Virus Res 60: 397–419.
- Rothman AL (2011) Immunity to dengue virus: a tale of original antigenic sin and tropical cytokine storms. Nat Rev Immunol 11(8): 532–543.
- Srikiatkhachorn A, Green S (2010) Markers of dengue disease severity. Curr Top Microbiol Immunol 338: 67–82.
- Guzmán MG, Kouri GP, Bravo J, Soler M, Vazquez S, et al. (1990) Dengue hemorrhagic fever in Cuba, 1981: a retrospective sero-epidemiologic study. Am J Trop Med Hyg 42: 179–184.
- Thein S, Aung MM, Shwe TN, Aye M, Zaw A, et al. (1997) Risk factors in dengue shock syndrome. Am J Trop Med Hyg 56: 566–572.
 Watts DM, Porter KR, Putvatana P, Vasquez B, Calampa C, et al. (1999)
- Watts DM, Porter KR, Putvatana P, Vasquez B, Calampa C, et al. (1999) Failure of secondary infection with American genotype dengue 2 to cause dengue hemorrhagic fever. Lancet 354: 1431–1434.
- Burke DF, Nisalak A, Johnson DE, Scott RM (1988) A prospective study of dengue infections in Bangkok. Am J Trop Med Hyg 38: 172–180.
- Guzmán MG, Kouri GP, Bravo J, Soler M, Vazquez S, et al. (1984) Dengue hemorrhagic fever in Cuba. II. Clinical investigations. Trans R Soc Trop Med Hyg 78: 239–241.
- Guzmán MG, Kouri GP, Bravo J, Valdes L, Vazquez S, et al. (2002) Effect of age on outcome of secondary dengue 2 infections. Int J Infect Dis 6: 118–124.
- Ooi EE, Goh KT, Wang DNC (2003) Effect of increasing age on the trend of dengue and dengue hemorrhagic fever in Singapore. Int J Infect Dis 7: 231–232.
- Lee MS, Hwang KP, Chen TC, Lu PL, Chen TP (2006) Clinical characteristics of dengue and dengue hemorrhagic fever in a medical center of southern Taiwan during the 2002 epidemic. J Microbiol Immunol Infect 39(2): 121–129.
- Figueiredo MAA, Rodrigues LC, Barreto ML, Lima JWO, Costa MCN, et al. (2010) Allergies and diabetes as risk factors for dengue hemorrhagic fever: results of a case control study. PLoS Negl Trop Dis 4(6): e699.
- Bravo J, Guzmán MG, Kouri GP (1987) Why dengue haemorrhagic fever in Cuba? 1. Individual risk factors for dengue haemorrhagic fever/dengue shock syndrome (DHF/DSS). Trans R Soc Trop Med Hyg 81(5): 816–820.
- Cunha RV, Schatzmayr HG, Miagostovich MP, Barbosa AM, Paiva FG, et al. (1999) Dengue epidemic in the State of Rio Grande do Norte, Brazil, in 1997. Trans R Soc Trop Med Hyg 93(3): 247–249.
- González D, Castro OE, Kouri G, Perez J, Martinez E, et al. (2005) Classical dengue hemorrhagic fever resulting from two dengue infections spaced 20 years or more apart: Havana, Dengue 3 epidemic, 2001–2002. Int J Infect Dis 9(5): 280–285.
- Lee KS, Lai YL, Lo S, Barkham T, Aw P, et al. (2010) Dengue virus surveillance for early warning, Singapore. Emerg Infect Dis 16(5): 846–849.
- Blacksell SD, Newton PN, Bell D, Kelley J, Mammen MP, Jr., et al. (2006) The comparative accuracy of 8 commercial rapid immunochromatographic assays for the diagnosis of acute dengue virus infection. Clin Infect Dis 42: 1127–1134.
- Cuzzubbo AJ, Endy TP, Nisalak A, Kalayanarooj S, Vaughn DW, et al. (2001) Use of recombinant envelope proteins for serological diagnosis of dengue virus infection in an immunochromatographic assay. Clin Diagn Lab Immunol 8: 1150–1155.
- Barkham TM, Chung YK, Tang KF, Ooi EE (2006) The performance of RT-PCR compared with a rapid serological assay for acute dengue fever in a diagnostic laboratory. Trans R Soc Trop Med Hyg 100: 142–148.
- Fried JR, Gibbons RV, Kalayanarooj S, Thomas SJ, Srikiatkhachorn A, et al. (2010) Serotype-specific differences in the risk of dengue hemorrhagic fever: an

AS VJL DCL. Contributed reagents/materials/analysis tools: JP AS VJL MLH KSC YSL DCL. Wrote the paper: JP AS VJL MLH KSC YSL DCL.

analysis of data collected in Bangkok, Thailand from 1994 to 2006. PLoS Negl Trop Dis 2;4(3): e617.

- Lee VJ, Lye DC, Sun Y, Fernandez G, Ong A, et al. (2008) Predictive value of simple clinical and laboratory variables for dengue hemorrhagic fever in adults. J Clin Virol 42: 34–39.
- Anders KL, Nguyet NM, Chau NV, Hung NT, Thuy TT, et al. (2011) Epidemiological factors associated with dengue shock syndrome and mortality in hospitalized dengue patients in Ho Chi Minh City, Vietnam. Am J Trop Med Hyg 84(1): 127–134.
- Ooi EE, Hart TJ, Tan HC, Chan SH (2001) Dengue seroepidemiology in Singapore. Lancet 357(9257): 685–686.
- Lye DC, Lee VJ, Sun Y, Leo YS (2010) The benign nature of acute dengue infection in hospitalized older adults in Singapore. Int J Infect Dis 14(5): e410-e413.
- World Health Organization (2005) Preventing chronic diseases: a vital investment: WHO global report. Available: http://www.who.int/chp/ chronic_disease_report/full_report.pdf. Accessed 2011 Nov 22.
- World Health Organization (2010) Global status report on noncommunicable disease. Available: http://www.who.int/nmh/publications/ncd_report_ chapter1.pdf. Accessed 2011 Nov 22.
- Geneau R, Stuckler D, Stachenco S, McKee M, Ebrahim S, et al. (2010) Raising the priority of preventing chronic diseases: a political process. Lancet 376: 1689–1698.
- Booth CM, Matukas LM, Tomlinson GA, Rachlis AR, Rose DB, et al. (2003) Clinical features and short-term outcomes of 144 patients with SARS in the greater Toronto area. JAMA 289(21): 2801–2809.
- Lau EH, Hsiung CA, Cowling BJ, Chen CH, Ho LM, et al. (2010) A comparative epidemiologic analysis of SARS in Hong Kong, Beijing and Taiwan. BMC Infect Dis 10: 50.
- Jain S, Kamimoto L, Bramley AM, Schmitz AM, Benoit SR, et al. (2009) Hospitalized patients with 2009 H1N1 influenza in the United States. N Engl J Med 361: 1935–1944.
- Dooley KE, Chaisson RE (2009) Tuberculosis and diabetes mellitus: Convergence of two epidemics. Lancet Infect Dis 9: 737–744.
- Dooley KE, Tang T, Golub JE, Dorman SE, Cronin W (2009) Impact of diabetes mellitus on treatment outcomes of patients with active tuberculosis. Am J Trop Med Hyg 80: 634–640.
- Charlton MR, Pockros PJ, Harrison SA (2006) Impact of obesity on treatment of chronic hepatitis C. Hepatology 43: 1177–1186.
- Jackson AL (2005) Evaluating diabetes mellitus as a risk factor for communityacquired infections. Clin Infect Dis 41(3): 289–290.
- Muller LM, Gorter KJ, Hak E, Goudzwaard WL, Schellevis FG, et al. (2005) Increased risk of common infections in patients with type 1 and type 2 diabetes mellitus. Clin Infect Dis 41(3): 281–288.
- Kaye WA, Adri MN, Soeldner JS, Rabinowe SL, Kaldany A, et al. (1986) Acquired defect in interleukin-2 production in patients with type I diabetes mellitus. N Engl J Med 315: 920–924.
- Geerlings SE, Hoepelman AI (1999) Immune dysfunction in patients with diabetes mellitus (DM). FEMS Immunol Med Microbiol 26: 259–265.
- Hsueh WA, Lyon CJ, Quiñones MJ (2004) Insulin resistance and the endothelium. Am J Med 117: 109–117.
- Dandona P, Alijada A, Chaudhuri A, Mohanty P (2004) Endothelial dysfunction, inflammation and diabetes. Rev Endocr Metab Dis 5: 189–197.
- Gonzalez-Curiel I, Castañeda-Delgado J, Lopez-Lopez N, Araujo Z, Hernandez-Pando R, et al. (2011) Differential expression of antimicrobial peptides in active and latent tuberculosis and its relationship with diabetes mellitus. Hum Immunol 72(8): 656–662.
- 45. Thomas L, Verlaeten O, Cabié A, Kaidomar S, Moravie V, et al. (2008) Influence of the dengue serotype, previous dengue Infection, and plasma viral load on clinical presentation and outcome during a dengue-2 and dengue-4 coepidemic. Am J Trop Med Hyg 78(6): 990–998.
- Gubler DJ (1998) Dengue and dengue hemorrhagic fever. Clin Microbiol Rev 11(3): 480–496.